Author's response to reviews

Title: Provision of inpatient rehabilitation and challenges experienced with participation post discharge: Quantitative and Qualitative inquiry of African stroke patients.

Authors:

Anthea J Rhoda (arhoda@uwc.ac.za)
Natalie N Cunningham (natalielingham@yahoo.com)
Simon S Azaria (simonazaria@yahoo.com)
Gerard G Urimubenshi (ugerard@khi.ac.rw)

Version: 5  Date: 16 July 2015

Author's response to reviews:

Dear Reviewers

We have made the recommended changes as requested. The word process was however replaced with provision instead of availability, as we thought this would be more appropriate. It could be argued that we did not investigate the availability of services.

Kind Regards
Anthea Rhoda